SPS STATUS (30/04 – 28/05)

· Ready for beam on 08/05

· Started setting-up of the LHC beam the same day on a 26 GeV plateau (16.8 s long).

· Running until 21/05 06:00 with LHC beam for conditioning of the vacuum chamber to increase the threshold intensity for beam induced multipacting (electron cloud) + studies with COLDEX apparatus

· Managed to inject up to 3 batches with more than the nominal intensity and kept within nominal transverse and longitudinal parameters at 26 GeV.

· 21/05 started setting up for fixed target physics at 400 GeV (North and West)

· Beam accelerated at 400 GeV by 17:15. reference measurements for alignment taken just in time (see below)

· 22/05: realignment

· SPS Setting-up completed by 24/05

· Physics started officially on 27/05 

Problems:

09/05 – Cooling BA1/BA3/BB3( BB3 compensator OFF (2:15) 

10/05 – 18 kV down in BA1: Transformer (1:30)

21/05 – Compensator 2 tripped (5:30)

22/05 – Compensator tripped (0:30)

22/05-23/05: cooling problem in the West area affecting some of the magnets in TT60 (extraction to the first TED) 

23/05 – 18 kV cell for SMD 7 broken (2:30)

24/05  TT20/TT60 power converters faulty (5:20) 

